Author's response to reviews

Title: Assessment of the Feasibility of High-Concentration Capsaicin Patches in the Pain Unit of a Tertiary Hospital for a Population of Mixed Refractory Peripheral Neuropathic Pain Syndromes in Non-Diabetic Patients

Authors:

Marc Gimenez-Mila (magimene@clinic.ub.es)
Sebastián Videla (svidela@esteve.es)
Marco- Antonio Navarro (dr_navarro_aguilar@hostmail.com)
Adela Faulí (afauli@clinic.ub.es)
Antonio Ojeda (ojeda@clinic.ub.es)
Ana Bogdanovich (abogdanovich@clinic.ub.es)
Luis-Alfonso Moreno (lamoreno@clinic.ub.es)
Clara Hernandez-Cera (chernanc@clinic.ub.es)
Carme Busquets (busquets@clinic.ub.es)

Version: 8 Date: 17 October 2014

Author's response to reviews: see over
Dear Dr. Tom Rowles,

Please find enclosed the revised version of the above mentioned manuscript. We have taken into account the comments made by the reviewers and the manuscript has been modified accordingly.

We enclose herein, as a supporting document, the list of changes that have been made. You will notice that the changes have been underlined.

We appreciate the thorough and relevant comments of the reviewers which, we believe, have permitted us to submit an improved manuscript. We hope that this revised version of our manuscript will be acceptable for publication in *BMC anesthesiology*, and we remain available for any further comments or suggestions.

Sincerely yours,

Marc Gimenez-Mila, MD
e-mail: MAGIMENE@clinic.ub.es
Editorial comment: We note that Table 1 again contains sufficient information to make the participants in your study potentially identifiable. We must therefore again request that you remove enough identifying information from this table so that it is not possible to identify the patients included in your study:

Answer: Some information from Table 1 has been removed just to avoid to identify the participants of the study.